We concentrate in this poster on two regions: the thermosphere (400-800 km, 7-150 nbar), and the stratosphere (80-280 km, 0.03-190 mbar). We can define a mean profile (solid line) by fitting a cubic spline, with spline points every 2 scale heights. The observed temperatures (dots) fluctuate around the mean profile.

Cautionary note: these data were current as of August 28 1996, but should be considered preliminary. Changes in the accelerometer calibration and different choices of the temperature at the highest altitude can affect the temperature and its first and second derivatives above 450 km. Fortunately, the derived fluctuations (raw-mean) are insensitive to all these changes below 800 km. So you can hear about observed waves now, but you’ll have to hear about the observed location of the heating at later meeting!

Mean temperature profile 2